

Received May 1, 2020, accepted June 8, 2020, date of publication June 16, 2020, date of current version June 26, 2020. *Digital Object Identifier* 10.1109/ACCESS.2020.3002793

# Adaptive Systems for Internet-Delivered Psychological Treatments

# SURESH KUMAR MUKHIYA<sup>©1</sup>, (Member, IEEE), JO DUGSTAD WAKE<sup>2</sup>, YAVUZ INAL<sup>3</sup>, AND YNGVE LAMO<sup>1</sup>

<sup>1</sup>Department of Computer Science, Electrical Engineering, and Mathematical Sciences, Western Norway University of Applied Sciences, 5063 Bergen, Norway <sup>2</sup>NORCE Norwegian Research Centre, 5008 Bergen, Norway

<sup>3</sup>Department of Information and Media Science, University of Bergen, 5007 Bergen, Norway

Corresponding author: Suresh Kumar Mukhiya (skmu@hvl.no)

This work was supported in part by the INTROducing Mental health through Adaptive Technology (INTROMAT) project through the Norwegian Research Council under Grant 259293/o70.

**ABSTRACT** Internet-Delivered Psychological Treatments (IDPT) are based on evidence-based psychological treatment models adjusted for interaction through the Internet. The use of Internet technologies has the potential to increase the availability of evidence-based mental health services for a far-reaching population with the use of fewer resources. Despite evidence that Internet Interventions can be effective means in mental health morbidities, most current IDPT systems are tunnel-based, inflexible, and non-interoperable. Hence it becomes essential to understand which elements of an Internet intervention contribute to effectiveness and treatment outcomes. By analogy, adaptation is a central aspect of successful face-to-face mental health therapy. Adaptability to patient needs can be regarded as an essential outcome factor in online systems for mental health interventions as well. While some aspects of rule-based and machine-learning-based adaptation have attracted attention in recent IDPT development, systematic reporting of core components, dimensions of adaptiveness, information architecture, and strategies for adaptation in the IDPT system are still lacking. To bridge this gap, we propose a model that shows how adaptive systems are represented in classical control theory and discuss how the model can be used to specify adaptive IDPT systems. Concerning the reference model, we outline the core components of adaptive IDPT systems, the main adaptive elements, dimensions of adaptiveness, information architecture applied to adaptive systems, and strategies used in the adaptation process. We also provide comprehensive guidelines on how to develop an adaptive IDPT system based on the Person-Based Approach.

**INDEX TERMS** Adaptive systems, Internet delivered psychological treatments, person based approach, information architecture, personalized Internet interventions, tailored Internet interventions, ICBT.

#### **I. INTRODUCTION**

In face-to-face therapy for mental health, therapists adapt their behavior towards their patients, to ensure the highest likelihood of successful therapeutic outcomes. The therapist's behavior is partly dynamic and can change in response to the patient's development. In Cognitive Behavioural Therapy (CBT) for social anxiety and fear of public speaking, for example, the therapist will begin with helping the patient to become comfortable with one particular social space. Once the patient is comfortable with that space, the therapist will adapt and help the patient generalize the progress and become

The associate editor coordinating the review of this manuscript and approving it for publication was Fabrizio Messina<sup>(b)</sup>.

comfortable in a broader range of social settings. In the case of reversal of development, the therapist will return to focusing on tasks previously managed by the patient. The therapeutic goals, dynamic guidance from the therapists, and the feedback that the patients experience with the therapists are essential outcome factors for mental health therapy.

Currently, online systems for mental health therapy are becoming widely available. Human beings are adept at understanding and communicating with others, and creating interpersonal contexts that are conducive to reaching a predetermined goal. These conditions are more challenging to re-create in online environments, however. It requires the re-alignment of roles, activities, and the remediation of information and intervention materials. While it is impossible to emulate face-to-face therapy in online settings completely, and online interventions are therapeutic constructions in their own right, one measure to help to ensure successful therapeutic outcomes is to make the environment adaptive to the user.

Building Adaptive Internet-Delivered Psychological Treatments (IDPT) requires input and competencies from both computer science and psychology, each presenting different challenges to building an adaptive system. From the computer science perspective, one challenge lies in the perpetual evolution of software systems and the integration of technology. As a result of this continuous evolution, software systems must be versatile, flexible, resilient, dependable, robust, energy-efficient, recoverable, configurable, customizable, and self-optimizing by adapting to operational changes in contexts, environments and system characteristics [1]. Additionally, the complexity and uncertainties of the problems arising in software systems are dynamic. These complexities prevent the use of a common a priori solution. Instead, an endeavor is made to solve this problem adaptively by dynamically collecting information about the problem at hand and using this information to generate an acceptable solution [2]. From the psychotherapy perspective, the ideal mental health treatment administered to one patient is not necessarily the same as that given to another patient exhibiting similar symptoms [3]. The individual nature of mental health problems makes it difficult to successfully make technology that resonates with human behavior and creates positive outcomes. People with similar mental health problems require different kinds of treatment. So, the technological solution needs to be adaptive to disseminate personalized treatments.

The Person-Based Approach (PBA) is a development methodology aimed at overcoming both Information and Communications Technology (ICT)-related and human-related challenges to build successful interventions for people to manage their health or illness [4]. PBA is explicitly developed for mental health interventions and health management and is meant to complement evidence, and theory-based treatment approaches by capturing and incorporating user perspectives in the design process [5]. PBA draws on methods from user-centered design [6], particularly behavioral analysis, and in-depth qualitative research [5], and seeks to understand both digital and non-digital behavioral aspects of interventions [7]. For mental health interventions to be effective, they must have a positive effect on the quality of life of the patient, they have to be technologically robust, flexible, and evidence-based, and they must have taken user interaction and engagement into account. Through building knowledge about user views, context and experience, the goal is to create interventions that are "maximally meaningful, feasible and engaging for all users" [7, p. 464], and ultimately to ensure the effectiveness of the intervention [5]. In this study, we further bridge the gap between approaches for participatory/user-centered and systems architecture design by adapting and expanding PBA to provide guidelines to build successful adaptive IDPT interventions. Adaptive systems for mental health therapy and PBA combine well, as both share the aim of increasing the usefulness for and relevance to the user. Furthermore, successful implementation of online mental health interventions requires the combination of several perspectives, including computer science and behavioral science. Hence, we consider the main contribution of our study to be a conceptual framework combining the systemic approach of model-driven software development of adaptive systems supporting online mental health therapy, with the empirically driven design framework of PBA. Both perspectives share the goal of facilitating the highest possible use of digital mental health interventions. We furthermore envisage that the resulting framework will allow developers, designers and researchers to reuse the same intervention system for multiple mental health issues.

We have tabulated a list of related works in Table 2 and compared them with our work in Section IX. To summarise, although some studies attempted to personalize interventions using the rule-based method and machine learning methods, none of the studies, to our knowledge, reported a reference model of an adaptive system and its core components systematically. We aim to fulfill this gap by proposing a reference model and outlining its core components.

The contributions of this study are the following:

- First, we propose a model that shows how adaptive systems are represented in classical control theory and show how the model can be used to specify adaptive IDPT systems. Moreover, concerning the reference model, we outline the core components of adaptive IDPT systems, the main adaptive elements, dimensions of adaptiveness, information architecture applied in adaptive systems, and finally, several strategies used in the adaptation process.
- Second, we provide comprehensive guidelines on how to develop an adaptive IDPT system building on the Person-Based Approach.

The rest of the paper is structured as follows: Section II outlines a definition of adaptive systems. Section III discusses adaptive IDPT systems and their types. Section IV describes several dimensions of adaptive IDPT systems. Section V outlines elements that can be adapted in an IDPT systems. Section VI describes different information architectures that can be observed in IDPT systems. Section VII outlines strategies that are used in the adaptation of IDPT systems. Section IX outlines a list of related work where we compare and relate our work with previous studies. Finally, Section VIII provides a guideline on how to develop an adaptive IDPT system with respect to the Person-Based Approach.

### **II. ADAPTIVE SYSTEMS**

An *adaptive system* is a set of interacting or interdependent entities, real or abstract, forming an integrated system that changes its behavior in response to its environmental changes [8], [9]. The changes that occur in its behavior are relevant and directed towards the fulfillment of the

Dimension	Descriptions		
User preferences	Temporal preferences, content presentation preferences, lingual preferences,		
	based on interaction log analysis		
Goals of the interventions	Therapists oriented goals, patient-centric goals, which are characterised by		
	evolution, flexibility, duration, multiplicity and dependencies.		
Measures	Psychometric tests: (Depression (MADRAS, MDI, BDI), Generalized Anxiety Disorder (GAD-7),		
	Bipolar Disorder (Altman Self-Rating Mania Scale),		
	ADHD (Adult ADHD Self-Report Scale) and others.), user behavior analysis based on interaction data		
Adaption actor	Automatic, semi-automatic (algorithm + therapist) and manual (Therapist/Users)		
Adaptation strategies	Rule-based, Goal-based, Machine Learning-based, Hybrid approach		

#### TABLE 1. Different dimensions of adaptive Internet delivered interventions.



**FIGURE 1.** We propose an Adaptive IDPT model based on classical control theory. It illustrates IDPT systems as a complex function that works in an environment (E), consumes controlled inputs (I), generates outputs (O). The system measures the performance of the outputs using some measures (P) and uses some feedback function (F) to create adaptive strategies (S). Actors (A) trigger the adaptation. These strategies control how the IDPT system should adapt.

objectives defined for the system. Self-adaptive software is context-dependent and has the ability to modify its structure and behavior in response to the contextual changes. This kind of software provides the users with context-dependent functionalities (presenting users with correct lingual interventions based on the country, for example – if a user is using IDPT system from any particular country, the intervention is presented in the language of that country) and context-independent functionalities (for example login form for authentication and authorization where different users credentials are checked for validity and permission with database). We chose to define adaptive IDPT abstractly, based on literature [10], [11], using classical control theory as illustrated in Figure 1.

The formal representation of the adaptive IDPT system constitutes the following components:

- a set of environments **E** in which IDPT is working as a complex process.
- a set of controlled inputs I consumed by IDPT systems for changing the behavior of the process. These inputs can vary from text-based inputs, sensor-based data such as heart rate, electro-dermal activities (EDA), user activities data, sleep data, voice data, eye movement data, and other types of biological markers.

- a process to measure the performance of the system **P** indicating performance of the IDPT in environment **E** when consuming **I** as input. Psychometric tests (see Table 1) are the principal performance measures for IDPT systems.
- a feedback function **F**, which generates an adaptive strategy with dynamic information about the process being controlled.
- a set of strategies **S** (see section VII), which use the knowledge learned from information and performance of the system.
- a set of actors A (see Figure 3) who trigger adaptation.

In general, adaptive software systems include monitoring, decision making, and actuation to maintain dependable behavior despite sudden, unpredictable changes like application workload fluctuations and hardware failures. As depicted in Figure 1, a set of inputs is monitored by an adaptive IDPT system to produce some output. Several performance measures, like psychometric tests and goal achievements, are used to evaluate the efficacy of the results (monitoring). Based on these measures, several strategies (decision-making process) are used to change the behavior of the system (actuation).

### **III. IDPT SYSTEMS**

Internet-Delivered Psychological Treatment (IDPT) systems refer to any software applications that facilitate interaction with psychological therapy through the Internet. These mainly include web-applications, mobile applications, augmented reality, and virtual reality applications. Several terms are used in research to refer to IDPT systems, for example web-based treatments, web-based interventions, online treatments, computerized psychotherapy, e-therapy, Internet-based cognitive behavioral therapy (ICBT or iCBT), digital interventions, internet-based interventions, online interventions, web-application based psychotherapy treatments, therapeutic web-based interventions, eHealth interventions [12], and others. In addition to this, researchers use different technical terms to represent the IDPT systems, for example Interapy [13], Deprexis [14], ULTEMAT [15], dBCIs (digital behaviour change interventions) [16], smartphone-based applications with specific brand names [17]. To be consistent, we chose to use the term



FIGURE 2. The conceptual model of Internet-Delivered Psychological Treatments (IDPT) includes one to many cases (Depression, Social Anxiety, and other mental health issues.). Each case contains one to many modules (Sleeping issues, Speaking issues, improving concentration, and others.) Each module, in turn, includes one to many tasks, which has prerequisites as the constraints. Each task contributes to collecting passive or active data in the system. Based on what type of data these tasks can accumulate, they consist of text task, audio task, video, exercises, or feedback.

*Internet Delivered Psychological Treatments*, as suggested by Andersson *et al.* [18].

IDPT has surfaced and grown as one of the most commonly practiced and widely researched forms of psychotherapy [19]. The evolution of IDPT, coupled with the exponential growth of Internet access throughout the world, has the potential to reshape the landscape of mental healthcare. Despite evolution in IDPT, several patients suffering from mental health issues go untreated [20], [21]. Obstacles to receiving treatment for mental health problems include long waiting lists, limited access to psychiatric medications, perceived stigma of seeking help, and treatment costs [21]–[23]. IDPT systems have been proposed as one solution to bridge this treatment gap. IDPT removes several barriers over traditional face-to-face therapy that hinders the majority of patients from efficient psychiatric care [24]. The use of IDPT tools can enhance patient health in several ways: 1) IDPT can be available and accessible from anywhere with an Internet connection [3]; 2) the temporal aspects of access can be substantially improved; 3) the scalability of IDPT can drastically enhance the functional capacity of the care; 4) makes the treatment cost-effective for individuals who do not have insurance or can not afford the out-of-pocket fees for treatment, and 5) removes the discomfort and the stigma related issues associated with the face-to-face approaches.

### A. COMPONENTS OF IDPT

Figure 2 depicts a conceptual model of IDPT. A typical IDPT contains several components:

- **Cases**: Typically, IDPT includes one or more cases such as Depression, Social Anxiety, Bipolar Disorder, ADHD, and other mental health issues [25].
- **Modules**: Each case contains one to many modules that focus on any particular dimension of the case. For example, for depression, there can be a module for sleeping issues, concentration issues, speaking issues, and others. One module can belong to one or many cases. The modules can have dependencies.
- Tasks: Each module, in turn, includes one to many tasks and have constraints as the prerequisites. Examples of such constraints include task dependencies, task availability, publication date, and others. Each task contributes to collecting passive (informative) or active (interactive) data in the system. Informative tasks provide learning materials about the mental health issue (case), symptoms, use cases, and several ways to manage them. The main objective of such educational materials is to provide psycho-education so that a) patients and their families can learn about symptoms, causes, and treatment concepts; b) patients can comprehend self-help program and steps required to manage their illness; c) patients can correlate their situations with similar others which helps to ventilate their frustrations. Such educational materials are in the form of reading tasks (text), listening (audio), and watching (video). In contrast to informative tasks, interactive tasks involve user interactions and often in the form of exercises. The exercises can be physical activities or computerized

tasks. Examples of activities include physical workouts and mindfulness exercises such as breathing exercises, walking certain distances, stretching, or physically performing any other activities. Examples of the computerized exercises involve fill in the blanks, question answering (Q/A), multiple-choice questions (MCQ), and feedback. The feedback type of task consists of using free text, rating systems, or multiple-choice questions.

## **B. TYPES OF IDPT**

- Guided vs unguided: IDPT can be either guided [26] or unguided [27]. In guided IDPT, a therapist is a planned part of the online intervention environment, and available to the patient to some degree. In unguided online therapy, the intervention is automated and independent of human support [28]. Several review studies (e.g., [28], [29]) find that participants in guided online interventions overall benefit more from the intervention compared to unguided interventions. The qualification of the therapist in these settings affects the patient results to a lesser degree [28]. A therapist in guided IDPT can make a diagnosis and determine the suitability of the treatment for a patient. The adherence rates in guided IDPT are also higher [29]. Several authors (e.g., [28], [30], [31]) have also found that the presence of accountable and trusted moderators, for example, enhances participant motivation and adherence in online therapy environments. There are, however, potential reasons for patients to prefer unguided options, such as privacy and increased likelihood of self-disclosure in stigmatic cases, and there are also ways of making unguided online intervention environments adaptive, such as automated notifications [32] and pre-intervention screening and tailoring [33].
- **Personalized care vs. stepped care**: There are two types of mental health care:
  - Personalized Care: Personalized medicine in P4 medicine [34] holds a great promise in improving the quality of treatments for mental health conditions [35]. The term P4 medicine stands for personalized (the focus of care is on the individual), predictive (signs of the illness can be recognized before it manifests), preventive (with correct tools and pre-diagnosis, can be prevented), and participatory (informed about the health conditions and healthcare decisions). With personalized care, treatment outcomes are improved by giving individual patients a treatment that is precisely right for them [36]. Personalized medicine is usually thought of as a way to match patients to the optimal intervention before initiating treatment [36], [37].
  - *Stepped Care:* In stepped care, all or most of the patients are given the same, least intensive, and most readily available treatment [38]. In stepped care, the treatments initiate with simpler form and

are gradually stepped up for those who can not benefit from simpler treatments [38].

### **IV. DIMENSIONS OF ADAPTIVE IDPT**

The way an adaptive system changes its behavior depends on a multitude of factors: a) users' preferences, b) goals of the interventions, c) measures, d) adaptation actors, and e) adaptation strategies. We refer to these aspects as the *dimensions* of adaptive IDPT. Andersson *et al.* [39] propose a classification of modeling dimensions for the self-adaptive software system. According to the study, the authors distinguish four dimensions of self-adaptive systems: *goals of the system, change that causes adaptation, mechanism which is the reaction of the system towards the change* and finally *effects which is the impact of the adaptation upon the system*. Similar to these dimensions, we propose five distinct dimensions as illustrated in Table 1 in the IDPT systems.

## A. USER PREFERENCES

Most of the mental health interventions have been modeled on a psychological therapy format with variation in the intervention period, content types, mode of delivery, and health issues. In general, the content is structured and designed to allow users to learn and practice new skills that help them to manage their illness. However, Internet content is often not designed in this way. Arguably, this is one of the reasons users spend less than 70 seconds on 80% of IDPT web applications [40]. It is reasonable to infer that users of IDPT have preferences for different models of engagement. User preferences [41] may influence other aspects of design, the modality of content, choice of videos, images, texts, or a combination of these that impact user engagement with online content [42]. User preferences can be the temporal, content presentation, and lingual. In temporal preferences, users tend to engage with interventions at different time instances of a day. In the content presentation preferences, the contents can vary from texts (reading), audio (listening), videos (watching) and exercises (writing). Users have different preferences regarding the type of content they like to interact with. In the lingual preferences, the users prefer to interact with the system in their language.

User preferences can be captured in two ways: explicitly asking and implicitly inferring, by recording the actions of the users as the form of a log. The latter choice is less intrusive. Log analysis helps therapists to understand their patients' preferences on the one hand and shows the bottlenecks in the system on the other. Agosti and Di Nunzio [43] presents a general methodology for gathering and mining information from Weblogs.

## **B. GOALS OF THE INTERVENTIONS**

Goals are the primary objective of the entities under consideration. Here, we outline two major types of goals: *patients goals* and *therapist goals*. In general, the goal of patients is to learn to manage their illness where the therapists' goal is to increase treatment outcomes so that the patients get better treatments. Goals of IDPT can have the following characteristics:

- Evolution: Goals can change within the intervention lifetime.
- Flexibility: Goals can be flexible in the way they are expressed. For example, *users must complete module one before going to module two* indicates a rigid goal. *A user can choose between assignments from the first module*, indicates an unconstrained goal.
- Duration: Goals of the intervention may hold validity throughout the intervention's lifetime or may be temporary.
- Multiplicity: The number of goals associated with the adaptive aspects of IDPT. An IDPT can have a single goal or multiple goals. Single goal IDPT systems are easier to implement.
- Dependency: In any IDPT system, goals can be related to each other. The goals can be dependent and independent.

## C. MEASURES

As illustrated in Figure 1, an adaptive IDPT system contains a way to capture user behavior. Behavioural data are essential inputs to an IDPT system, to make it possible to adapt to the user in a meaningful way. Several data sources can be used as relevant indicators of patient behavior and symptom development. In guided online interventions, self-assessment using validated screening tools, telephone, video conversations, and written communications can be used. Counselors in guided IDPT also assess written responses to assignments and patient diaries. Additionally, also for non-guided interventions, it is possible to use indicators such as system interaction data to gauge patient engagement with an intervention. The potential of using sensors embedded in smartwatches, mobiles, and other devices has also been explored, for example, to measure physical mobility, sleep patterns, voice, medication adherence, and heart rate variability, to name a few. Here we consider the measures of psychometric tests and user behavior analysis:

- *Psychometric tests*: Psychologists have developed an effort to quantify people's preferences, behavior, and intelligence through self-assessment questionnaires referred to as psychometric tests. For example *PHQ-9 [44], MADRS-S [45], and BDI [46]* is used for assessment of depression. *GAD-7* [47] is used in self-assessment of general anxiety disorder. Scores from these tests can be utilized to adapt the modalities of content.
- User behavior analysis: We can analyze user behavior based on the interaction between the user and an adaptive IDPT system to gather user preferences and learn what a user likes the most. These insights can be extracted by analyzing the usage intensity of the system, such as the time a patient spent on watching videos, listening to audios, reading texts, interacting with intervention exercises, and other activities. The analysis of intensity can reveal user preferences and the

engagement of the patients [48]. We can use the preferences information to personalize the adaptive elements (see Section V). Moreover, therapists can compare the expected behavior of the user and the actual behavior. This comparison can reveal where the bottlenecks are in the system and guide us to improve the structure of an IDPT system.

# D. ADAPTATION ACTORS

There are three different types of adaptive systems based on the actors involved in triggering adaptation. First, the automatic systems that have the capability for self-adaptation. Second, the semi-automatic systems where a portion of adaptation is done by algorithms and a portion of adaptation is controlled by the therapists. Finally, manual systems where an actor (therapist/medical practitioners/trained medical personnel) manually adapts the interventions in IDPT. The use case diagram, as Figure 3 illustrates different types of actors involved in an adaptive IDPT system. The actors adaptive systems and medical practitioners are actively involved in the task of triggering/creating an adaptation. The actor patients uses the adaptive system, and finally, the actor IT admin monitors and maintains the adaptive system to ensure proper functioning. We like to highlight that the use cases functionalities illustrated in Figure 3 provide a generic example and can differ according to the context of the project.

## E. ADAPTIVE STRATEGIES

Adaptive strategies refer to a set of techniques applied to change the behavior of an adaptive system. In Section VII, we outline four different types of adaptive strategies including *rule-based, goal-based, feedback-loop based* and *ML-based* approaches.

## V. ADAPTIVE ELEMENTS: WHAT CAN BE ADAPTED

Having argued for the necessity of adaptation and different dimensions of the adaptive system, in this Section, we identify what can be tailored in IDPT. Although there is no explicit study done, to our knowledge, to categorize what can be adapted, we present a list of elements in IDPT that can be adjusted.

- *Content presentation*: IDPT systems use psychoeducation as the primary technique to educate patients on any healthcare problems. To do so, therapists prepare several educational materials such as use cases, events, activities, and physical exercises to familiarize patients with the symptoms of any particular illness. We refer to these educational materials as intervention content. We can present these contents in several modalities (e.g. slides, audio, video, animations, and texts) and layouts in any IDPT system. How we present the content to the user can be adapted based on several dimensions, as shown in Section IV.
- *Information architecture*: Section VI presents four different types of Information Architecture (IA) that can be adapted based on various dimensions.



**FIGURE 3.** The Figure illustrates the different types of actors involved in adaptive IDPT systems. The actors' *adaptive systems* and *medical practitioners* (top block) are actively involved in the task of triggering/creating adaptation. The actor *patients* (bottom block) use the adaptive system, and finally, the actor *IT admin* (bottom block) monitors and maintains the adaptive system to ensure proper functioning. The dots illustrate the actors can perform several other actions defined by the context of any project.

- *Content complexity*: Content complexity is associated with cognitive demands inferred from the language of a content standard. In essence, content complexity incorporates several factors such as prior knowledge, processing of concepts, skill set, sophistication, application of content structure, number of parts, and length of the content. Because of its dependency on prior knowledge and processing skills that vary between person to person, content complexity adaptation based on several dimensions is essential.
- *Content recommendation*: If an IDPT is recommending content to the participants, it should be adapted based on the user preferences, interests, and other dimensions mentioned in Section IV.
- User Interface: Adaptive User Interface (AUI) changes its layouts and its elements according to the need of the user or the context [49]. Often, creating AUI is primarily based on the features of IDPT systems, and the knowledge levels of the users that will use the systems. By making these adaptations, the IDPT systems gain an ability to conform to the users' needs. For example, *adaptive presentation* aims to display certain information based on the current user. If the current user

indicates novice knowledge, the system can adapt to display minimal information. Conversely, if a user shows advanced understanding, the system adapts to illustrate with more detailed information. Using adaptive strategies is likely to overwhelm less [50], [51] novice users with complex content. Similarly, *adaptive navigation* intends to guide a user to their specific goal within the system by transforming the way the system is navigated based on certain factors (like goals of the user, previous history information, current skill levels) of the user.

- *Feedback*: Several studies indicate that incorporating feedback mechanisms increases user adherence. Several forms of feedback mechanism has been used including *email, SMS, phone calls, conversational agents (chatbots [41])* and *support forms*. However, there can be several perspectives on how patients perceive feedback. Hence, it should be adapted based on the patients' needs.
- *Notifications/Alerts/reminders*: In an information system, personalized information is sent to patients as reminders, alerts, or notifications. Notification is a technique of bringing something to notice to the users. Reminders are specific types of notifications that are sent to stress about upcoming events. In an IDPT system, reminders have been used to notify clinicians and patients about appointments, screenings, upcoming interventions, tasks, or activities. An alert is a type of indicator such as a badge, sounds, missed required fields, success, or fail messages to inform users that they should look at something. Unlike alerts, a notification is accompanied by more information and is not always in the current task context.
- *Reports*: Reports are reflections of data collected by IDPT that are sent back to the users. Reports may include a subset of visualization techniques conveying specific information generated from the previous state of the users and current assessment. For example, a graph showing the current calorie count of the users, thoughts records.

### **VI. INFORMATION ARCHITECTURE OF CURRENT IDPT**

Information Architecture (IA) plays a vital role in web application development, and a good architecture can enhance the ability of employees and customers to find information and decrease the application maintenance cost [52]. Despite some recommendations for using Architecture Centric Development (ACD) in behavioral and mental health interventions, there is still a lack of empirical evidence for the role of architecture in IDPT [50]. Current IDPT using ACD primarily focus on navigation systems [53] and their architecture can be *hierarchical, matrix, hybrid* and *tunnel based* [50], [54].

 Tunnel design: 90% of the available IDPT systems use tunnel-based design [55] where users navigate sequentially through the intervention. Tunnel based design is analogous to watching TV, reading textbooks, attending academic classes, or attending multiple clinical sessions. An argument for this model is that a tunnel design experience is less likely to overwhelm users with information and options [50], [51].

- 2) Matrix design: Matrix IA is designed on the principle of hypertext HTML. In a matrix design, the user can freely pursue their interests by using their path available through information. The designers of matrix IA use several organizational schemes to group the content into several categories. Examples of organizational schemes include alphabetical order, chronological order, topic scheme, and audience scheme. The primary rationale of using the matrix design is to provide maximum information within a system. According to Lynch and Horton [56], matrix design may not be well-suited to helping users to become familiar with new content. However, it is more applicable to web applications that are designed for highly educated and experienced users.
- 3) **Hierarchical design**: Information is organized in a top-down manner so that users can review specific content in a non-sequential way. The hierarchical design provides a simplified view of the content, making it easy to retrace steps. However, deeply nested information may be difficult to locate [57].

As outlined in Subsection, III-A, we view an adaptive IDPT to follow the hierarchical Information Architecture. A common representation of the hierarchical IA is in the form of a graph. Here, a graph can illustrate the components of an IDPT system. Formally, we define a graph G as a pair (V, E) where V is a set of vertices, and E is a set of edges between the vertices  $E \subseteq (u, v)|u, v \in V$  [58]. The vertices of the graph represent the components of an IDPT system, and the edge represents the constraints. For example, a module in an IDPT system contains several tasks. Each task represents the vertices. A patient must fulfill one or more constraints before we mark the task as completed by the user. We indicate these constraints by edges. For example, task dependencies constraint indicates the patients must complete a certain task before other tasks are available. Formalizing an IDPT system using graph theory is beyond the scope of this paper, and is marked as one of the immediate future work.

4) Hybrid design: It is a combination of one or more IA designs that best-fit content and purpose. Several current web application takes advantage of hybrid design [57] as it combines merits from other models.

It is not straightforward to know what components make up an Information Architecture [52] because people directly interact with some components. In contrast, other components are behind the scenes that people are not aware of their existence. In general, any IA consists of four different components:

1) *Organization systems:* It deals with how we categorize the information, such as by subject, by chronological

order, by alphabetical order, by geographical order, by audience types, and so on. Organizing information is crucial because it reduces ambiguity in finding information, reduces heterogeneity, and labeling systems are intensely affected by the creators' perspectives. The difference in attitude makes information organization a problematic task.

- 2) *Labeling systems:* Labeling deals with the form of representation. A label illustrates a chunk of information. For example, Frequently Asked Questions (FAQ) is a label that describes a piece of content that most of the users typically wonder. Labeling is crucial as it 1) communicates with the patients revealing them the essential information, and 2) the interaction data with labels help in using supervised machine learning algorithms [59].
- 3) Navigation systems: It deals with how we allow users to browse or move through the information in a system. The navigation system facilitates browsing and contains several types, such as global navigation, local navigation, and contextual navigation. A global navigation system is present on every page throughout the application. Local navigation helps to browse on a specific page. For example, often, we provide an index of the main sections in any application page. Contextual navigation aids providing contextual navigation links that are specific to any particular piece of information. Such contextual navigations are applicable when we intend to provide extra contextual information if the user intends to. For example, in an adaptive IDPT system, we offer tasks that facilitate psycho-education related to any particular symptoms of mental health. However, we do not discuss in detail about the symptoms. In such a case, we can provide contextual navigation by using a hyperlink that links to a document providing detailed information about the symptoms. Once we design the navigation system, we can adapt these based on the user search behavior. Gobert et al. [60] presents SAM, a modular and extensible JavaScript framework for self-adapting menus on the web pages. According to the study, SAM provides a mechanism to self-adapt the navigation systems, which in turn offers personalized web experience.
- 4) *Searching systems:* It deals with how users search for information. A searching system is a challenging and well-established field on its own. Hence, before investing time and budget on creating a search system, we recommend determining whether the product needs a search system. If we decide to build a search system, then it requires defining the basic anatomy of the search engine, what is searchable, choice of retrieval algorithms, understanding retrieval results, and the search interfaces.

Explaining all the components of IA in depth is beyond the scope of this paper. The book by Morville [52] outlines these components in detail.



**FIGURE 4.** In a rule-based adaptive strategy, a set of predefined rules (managed by the psychiatrists and referred to as rule engine) modifies the behavior of the system. When these rules are truthy, an adaptive actor triggers the adaptation.

## **VII. STRATEGIES FOR ADAPTATION**

In the last decade, several strategies for adaptation has been practiced to improve the effectiveness of IDPT, and a multitude of endeavors to generalize these strategies into discrete classes has been carried out. For example, Salehie and Tahvildari [61] identified two categories of self-adaptation based on impact and cost factors. Qureshi et al. [62] categorized four different categories of self-adaptive systems: Type 1 consists of systems that anticipate both changes and possible reactions at the design stage; Type 2 consists of systems that own many alternative strategies for reacting to changes; Type 3 consists of systems aware of its objectives and operating with uncertain knowledge about the environment and; Type 4 is inspired by biological systems that are able to self-modify their specifications when no other possible refinements are possible. Analogously, the paper [63] represents four similar types (like [62]) of self-adaptive systems using a metamodel. For IDPT systems, we categorize four different types of adaptive strategies: a) rule-based adaptation, b) adaptation through feedback loops, c) goal-driven adaptation and d) adaptation through predictive algorithms.

#### A. RULE-BASED ADAPTATION

As depicted in Figure 4, the rule-based adapting systems, changes its behavior based on pre-defined rules. The psychiatrists manage these rules based on their domain expertise. These rules are in the form IF <condition> THEN <action>. We can use the Horn clause [64] to formulate one or more pairs of the conditions. In an adaptive IDPT system, the logical structure of the clauses are of the form:

$$\langle C_1 \wedge C_2 \vee C_3 \vee \ldots \vee \neg C_n \rangle \implies A_1$$
 (1)

$$\langle C_{k1} \lor \neg C_{k2} \land C_{k3} \lor \ldots \lor C_{kn} \rangle \implies A_2$$
 (2)

where:

 $C_i =$ conditions

 $A_i$  = action if the condition is true



**FIGURE 5.** An example illustrating *rule based adaptation* where a therapist could create several alerts based on *psychometric tests score, user activity, goals and other* appropriate variables.

The quality of these rules and the quality of the performed adaptive actions are measured using several techniques [65] in the literature, including confidence value, rule weight, priority, check function [66], similarity degree, average fitness, case-based reasoning, system metrics and users' preferences [67]. Rule-based adaptation has advantages of readability, transparency, elegance, and they are easily modifiable [65]. According to the same survey, rule-based adaptation has been widely practiced in user interface adaptation [68], process adaptation [66], [69], workflow adaptation, content adaptation [70], software configuration adaptation, and feature adaptation in Figure 5, where an adaptive system sends a reminder (based on predefined email template) to a therapist if the following conditions are True:

- 1)  $(C_1)$  If the Psychometric score is GAD-7 for Generalized Anxiety Disorder and its value is greater 21 and,
- 2) ( $C_2$ ) The user has not logged in to the system for more than 72 hours ( $C_{2,1}$ ), or the user's goal is to decrease anxiety ( $C_{2,2}$ ). The condition  $C_2$  evaluates to false only if both  $C_{2,1}$  and  $C_{2,2}$  are false.

### **B. ADAPTATION THROUGH FEEDBACK LOOPS**

Feedback loops are regarded as important aspects of self-adapting software systems. Brun *et al.* [71] argue feedback loops as essential for building self-adaptive software systems and for software evolution. A generic feedback loop contains four steps: collection, analysis, decision point, and action [72]. Several sensors based systems are utilized to collect data about the adaptive system's current state and context. The accumulated data are then transformed (cleaned, filtered, pruned, stored) for future referenced. The decision step then analyzes the data to infer trends and identify key patterns. Finally, based on the analysis results, a set of actions are decided that are executed using context actuators.

### C. GOAL-DRIVEN ADAPTATION

In the goal-driven adaptation, the organizational and patient needs, in addition to their goals, are the main factors of adaptation. Heaven *et al.* [73] describes three-layer model controlling goal-driven adaptations- *goal management, change management and control.* The goal management layer



FIGURE 6. Adaptation based on predictive algorithms: Users' history data is fed to a suitable Machine Learning classifiers, and a model is trained. The users' current state (sensors data) is used as test data with the trained model to get possible predictions. Based on the prediction, a suitable treatment module can be used based on the pre-designed knowledge base.

handles the creation of a new set of goals or re-planning an existing set. The change management layer controls the adaptation among several tasks at hand based on the goals. Finally, the control layer is the level of execution of the selected configuration.

## D. ADAPTATION THROUGH PREDICTIVE ALGORITHMS

The recent rapid growth in Artificial Intelligence (AI) has opened up unprecedented possibilities in analyzing and predicting diverse phenomena, including humanities, social and cognitive sciences, finances, healthcare, robotics, and other areas of natural sciences. Machine Learning (ML), a subset of Artificial Intelligence (AI), provides principled methodologies for the development of automatic, complex, objective algorithmic models for the analysis of multi-dimensional and multi-modal biological data. For instance, speech data that can be used in automated detection of mental health morbidities with the application of ML approaches. A study was done by Garcia-Ceja et al. [59] outlines several studies done about mental health monitoring system that uses sensors and machine learning. Similarly, ML-based algorithms have been used to predict user adherence to IDPT for depression and anxiety after myocardial infarction [74]. The use of ML-based predictive algorithms can be used to increase adaptation in IDPT systems, and the overall process is illustrated in Figure 6.

As depicted in Figure 1, the predictive algorithms are a type of adaptive strategy. These algorithms generally learn complex rules from the users' historical data. We store the patients' history data and feed these data to the predictive algorithms to obtain a trained model. Once we get a trained model, we input any user's current (sensor data) state to the trained model, which provides a set of possible predictions. Based on these predictions, we can get possible treatments based on the intervention knowledge base (KB). Here, we assume based on the domain expert knowledge (therapists), we can design a knowledge base. For example, Generalised Social Anxiety (GAD-7) [47] distinguishes three classes of anxiety based on symptom severity - mild, moderate, and severe. So, a typical KB would contain three

different types of interventions for each category. When a user takes this intervention, we store their activities as a log file. We analyze this log file to create an understandable user profile. We refer to this process as user profiling.

## VIII. DEVELOPING ADAPTIVE IDPT SYSTEM BASED ON PBA

In the previous sections, we have outlined several components of adaptive IDPT and covered several aspects of developing adaptive IDPT systems. In this Section, we provide an account of how to design and develop an adaptive IDPT system building on the PBA approach. PBA is well suited to complement the systems architecture and development approach because of it's focus on user needs. Hence, we map the development of an adaptive IDPT system to the PBA approach and present guidelines on how to develop an adaptive IDPT system based on user perspectives and needs. Here we utilize the design scaffolds available in PBA to provide insights into how to build user-friendly and relevant adaptive IDPT systems. The PBA proposes the development of digital health interventions in four distinct stages as depicted in Figure 7; (1) planning, (2) design, (3) development and evaluation of acceptability and feasibility, and (4) implementation and trialing [4]. PBA is a methodology for developing health and illness management interventions, inspired by user-centered design, that has the goal of providing as much value as possible to the end-user. Here, we explore how to supplement and expand the person-based approach with guidelines on how to develop an adaptive IDPT system.

#### A. PLANNING

During the planning phase, it is essential to define the problem domain, its scope, context, and main challenges of current problem-solving systems or techniques. Having understood the problem domain, we plan a new system and determine its objectives. We then perform CBA (Cost-Benefit-Analysis) of the proposed system, determine its SWOT (Strength, Weakness, Opportunities, and Threats) features, and finally perform feasibility analysis. The feasibility analysis should focus on different perspectives:

- 1) *Technical feasibility*: It is recommended to consult technical expertise to determine if the development of the proposed IDPT system is technically feasible.
- 2) *Economical feasibility*: Financial feasibility is an essential parameter to determine the level and complexity of the adaptive IDPT system.
- 3) Legal and ethical feasibility: The proposed system must abide by the legal requirements of the region and concerned stakeholders. It is relevant to know if the system meets within the legal and ethical constraints of the business, organization, and demography.
- 4) *Operational feasibility*: It is essential to identify if the proposed system solves the problem space that is originally identified.
- 5) *Behavioral feasibility*: It is important to take user attitude, behavior, or contextual consent towards



FIGURE 7. In this model, we compare the development of an adaptive IDPT system with the Person-Based Approach (PBA) and provide comprehensive guidelines about how to develop an adaptive IDPT system.

the development of the new system. As aforementioned, in Section I, there is currently a sub-optimal patient adherence towards IDPT programs. Hence, it is recommended to take user behavior and their need for the development of adaptive IDPT systems.

6) *Schedule feasibility*: It is essential to know if the adaptive IDPT system can be developed in the allowed time and within the offered resource and budget.

Moreover, we require to work on the design of the user interface and discuss the user interface choices such as multi-touch sensors, gesture-based, Augmented reality (AR) and Virtual Reality (VR) technologies, web-applications, smartwatches, and others. We need to comprehend available technologies and how these technologies assist with adaptation and discuss how users interact with these systems. The problem space needs to be articulated from the perspectives of both computer science and psychotherapy since each has different challenges to develop an adaptive IDPT system. In addition to usability and user experience goals, goals for adaptation are necessary to be considered to define the problem space thoroughly, which results in transferring the knowledge to design space. The main questions that we need to address during planning phases are listed below:

- What is the problem space we are addressing?
- What are the dimensions of the adaptive system?
- What are the main elements we want to adapt?

Theories, previous quantitative research evidence, systematic reviews, meta-analyses can provide insight into the intervention components and perspective of people who will use the intervention. The planning phase of PBA helps in exploring and analyzing the attitudes, understanding, needs, and situations of people who will be interacting with the systems. The exploratory analysis helps to disclose the dimensions of adaptiveness, the choice of user interfaces, and the main adaptive elements required in the intervention. The preference for user interfaces varies according to gender, cultural background, health literacy, or previous experiences. Hence, intervention developers need to consider purposefully sample a diverse range of users who differ in the characteristics to understand their actual needs. Understanding these needs helps to ensure that the researcher has all the insight into all relevant perspectives, allowing to server tailored intervention based on their needs.

# B. DESIGN

During the design phase, we first need to define the requirements, evaluate the possible alternatives, and prioritize these requirements lists. Classifying these requirements into functional and non-functional requirements help to prioritize them.

- Functional requirements: Functional requirements for adaptive IDPT systems include secure authentication and authorization; the ability for therapists to create interventions and adapt based on several adaptive strategies (see Section VII); adaptive dashboard for both therapists and patients; administrative functions; and audit tracking. Moreover, these requirement lists can differ based on legal and organizational aspects.
- Non-functional requirements: Non-functional requirements includes product requirements (usability, dependability, efficiency requirements, interoperability requirements [75]), organizational requirements (development requirements, operational requirements), and external requirements (ethical, regulatory, and legislative requirements) [76].

Based on these requirements list, we need to create a Software Requirement Specification (SRS) [77]. In the design phase, we also need to create a design for the network, databases, user interfaces, and system interfaces. The designers transform the requirements from the SRS documents into the logical structure. These structures include:

- 1) *Architectural design*: In the architectural design, we identify the overall structure of the system (refer Subsection III-A), the main adaptive elements (refer Section V), their relationships, and their distribution.
- 2) *Database design*: Database design involves the creation of the system data structures and their relationships.
- 3) Interface and interaction design: Interface design requires the involvement of Interaction Design (IxD) experts to create interactive, precise, and unambiguous user interfaces. Interface design is also concerned with the specification of detail, including syntax and semantics of the services that are provided the user interface. Several structural diagrams can be used to model syntax of the interface, and the semantics of the interface may be defined by using constraint languages for example Object Constraint Language (OCL) [78].
- 4) Component selection and design: In this design, we need to search for reusable components. Since adaptive IDPT is mostly being developed as web or mobile-based apps, there is a large community of open-sourced and proprietary ingredients that helps in building rapid prototyping. Hence, instead of reinventing the well, we recommend identifying the usable components and utilize them for the creation of an IDPT system.

Lastly, we need to review the proposed design and ensure the final design adheres to SRS. The outcome of design phase is detailed *system architecture, database design, interface design,* and *component descriptions.* 

# C. DEVELOPMENT AND EVALUATION OF ACCEPTABILITY AND FEASIBILITY

In the development and evaluation phase, the developers transform the design specifications into source code, where several modules come together to perform an integrated system. This phase is also referred to as the *coding* phase and is the most prolonged phase of a development life cycle. It includes careful development planning, plan analysis, designing, actual coding, and testing.

- During the development planning phase, the development needs to decide appropriate programming languages, frameworks, databases, versioning systems, project management tools, and other tools required to develop the adaptive IDPT system. The process of selecting tools requires necessary feasibility analysis concerning business goals, available budgets, and schedules.
- The analysis involves making the detailed logical architecture of the applications, network, and database. An analyst decomposes the system into several parts called "Objects". Decomposition helps to identify reusable components increasing development efficiency and cost-effectiveness.
- The Designing phase is a creative step that involves making decisions about the system, for example which programming language to use, which database vendors to integrate, hardware, and network decisions.
- Coding is a mathematical approach that involves how logic operates. Coders follow specific styles, formats, and patterns to create the programs.
- Finally, the testing phase involves the verification and validation of the product. It consists of debugging by the programmers to ensure that the programs execute as intended. In addition to debugging, the product should need to go through the quality assurance (QA) process. QA involves acceptance testing that allows validating the software based on the original SRS. Acceptance testing involves several other forms of testing, including system testing (load testing, stress testing), integration testing, compatibility testing, user testing, and others [79].

## D. IMPLEMENTATION AND TRIALING

In this stage, we apply the adaptive IDPT system into practical use - that is, clinical trials and experimental evaluations. Evaluation of problem space and user needs, usability, and user experience is an essential aspect of the development of digital tools that should be performed during development. When a stable and usable system is achieved, measuring and building evidence of the medical effects of the intervention is equally important. The development of online interventions requires a combination of the scientific traditions of computer science and medicine. In practice, the traditions have different conceptions of what can be taken as evidence, and how to construct it, and requires the evidencing to take place over different temporal phases. Corrections can be done in a pilot

#### TABLE 2. List of related studies attempting to adapt IDPT systems.

Reference	Year	Objective	Main findings
Brian <i>et</i> <i>al.</i> [57]	2005	Reviews the roles of considering IA designs for effective web-based interventions	Encourage adoption of a multidisciplinary perspective IA for presenting content of behavior change interventions.
Webb <i>et al.</i> [82]	2010	Investigate which characteristics of IDPT best promote health behavior change	<ul> <li>a) IDPT incorporating more behavior change techniques (stress management, communication skills, problem-solving and others) have larger effects on outcomes. b) Additional communication with participants using SMS, email, messages increases adherence.</li> </ul>
Christensen <i>et al.</i> [83]	2009	Review adherence with respect to IDPT and investigate the rates of dropouts and compliance in RCT for anxiety and depression	Main predictors of adherence include disease severity, treatment length, and chronicity. Very few studies examined the actual reasons for dropout.
Andersson <i>et al.</i> [18]	2019	Literature review arguing ICBT can be viewed as a vehicle for innovation	Detailed review in several directions including effectiveness of IDPT, implementation paradigm, predictors and future works.
Kelson <i>et al.</i> [84]	2019	Examine the therapeutic impact of Internet-delivered acceptance and commitment therapy(iACT) on all anxiety conditions.	Results indicate iACT to be efficacious and acceptable treatments.
Jokste <i>et al.</i> [65]	2017	A systematic review of rule-based adaptation to discover types of rules applied, application domains, and performance measures.	<ul> <li>a) Results indicate widely adopted in medicine-related system, an adjustable system for people with disabilities and others. b) Categorized three forms of semantic rules: event-condition rules (if-else), association rules, and RuleML based behavior rules.</li> </ul>
Karyotaki <i>et al.</i> [85]	2006	Examines the predictors of dropout in an individual patient data meta-analysis	<ul> <li>a) Dropout can be predicted by several variables and is not randomly distributed.</li> <li>b) Understanding these variables can help to adapt IDPT to prevent dropout in identified groups at risk.</li> </ul>
Rogers <i>et al.</i> [86]	2017	<ul> <li>(a) discover the range of health-related topics that are addressed through Internet-delivered interventions, (b) generate a list of current websites used in the trials, and (c) identify gaps in the research that may have hindered dissemination.</li> </ul>	Wide range of IDPT is available for health-related behavior. However, most of the IDPT found to be efficacious in RCT does not have a website for general use.
Brouwer <i>et al.</i> [87]	2011	Identify (a) which potentially exposure-promoting methods and strategies are used in existing IDPT, b) which objective measures are used to measure exposure to IDPT, c) which methods are associated with better exposure.	Feedback, interactive elements, and email/phone contact was mostly used methods and strategies to increase treatment outcomes. No clear conclusion is drawn due to the diversity of intervention methods used and inconsistency in reporting.
Pugatch <i>et al.</i> [50]	2018	Synthesize the existing literature on website information architecture and its effect on health outcomes, behavioral outcomes, and website engagement.	The authors only found three relevant papers. The first paper that investigated IA exclusively found that a tunnel IA improved site engagement and behavior knowledge, but it decreased users' perceived efficiency. The second and third papers found that a tailored site condition improved site usage, behavior knowledge, and some behavior outcomes.

trial where the developed adaptive IDPT system is accessed by users who test if an intervention is technically working, the presence of logical errors, and other functionalities by initial requirement specifications. The testing of adaptability in a clinical trial requires an evaluation of all potential configurations of adaptive components. To measure the medical outcomes and effects of adaptability requires a comparison of adaptive systems to non-adaptive systems. Additionally, adaptive aspects or elements of a system can be compared to non-adaptiveness or each other. Currently, Randomized Controlled Trials (RCTs) are not well matched to the pace of technology development, and there is a need for methodologies that can account for effects more rapidly [80]. One recent methodological approach that is aimed at meeting evidence demands from medicine, simultaneously to being relevant to methodological requirements from technology development is found in micro-randomized trials [81] Micro-RCTs aims to provide a data-based method for evaluating online interventions by providing an experimental design for use in testing the proximal effects of the newly developed treatments.

## **IX. RELATED WORKS**

With the prevalence of mental health illness, on the one hand, the enhancement in ubiquitous computing on the other, several authors have attempted to adapt intervention in several ways with different objectives. Table 2 contains a list of similar studies done previously. The studies [50], [57] reviewed the roles of IA designs for effective web-based intervention. However, the studies are limited to IA and behavior changes. In particular, IA is just one of the characteristics of IDPT. Webb et al. [82] performed a systematic review to investigate which characteristics of IDPT best-promoted health behavior and reported IDPT that incorporated more behavior change techniques (stress management, communication skills, problem-solving and others) have more significant outcomes. However, it is still not sufficient to build an adaptive system only based on behavior change techniques, as several other dimensions (disease diversity, treatment length, chronicity) contribute to user adherence in the IDPT system [83].

Christensen *et al.* [83] performed a systematic review to investigate adherence to IDPT and the rates of dropouts for anxiety and depression. The study reports the main predictors of adherence are disease severity, treatment length, and chronicity. However, the study by Karyotaki *et al.* [85] reported that several variables could predict dropouts, and these variables are not randomly distributed. Understanding these variables can help to adapt IDPT and prevent dropouts in the identified groups as a risk. Adhering to this conclusion, we aim to present, in our study, several variables (adaptive dimensions) that affect adaptation.

Jokste and Grabis [65] conducted a systematic review of rule-based adaptation to discover types of rules, application domain, and performance measurement. The study by Brouwer *et al.* [87] attempts to identify which exposure-promoting methods and strategies are used in existing IDPT system and identify performance measures in the IDPT environment. The study reported that feedback, interactive elements, and email/phone contact were mostly used strategies. However, no clear conclusion was drawn due to the diversity of intervention methods used and inconsistency in reporting. While the study uses feedback, interactive elements, and email/phone contact as strategies, these are just the rule-based adaptation mechanism. Feedbacks, supports are the adaptive elements.

Rogers *et al.* [86] attempted to discover the range of healthrelated topics that are addressed through Internet-Delivered interventions and identify the gaps in the research that may have hindered dissemination. The study reported a wide range of IDPT systems is available for health-related intervention. However, most of the IDPT reported being efficacious in RCT trials does not have a website for general use. Hence, the internal architecture, adaptive elements, and strategies used remain unknown for the research community. Andersson *et al.* [18] performed a systematic review arguing ICBT can be viewed as a vehicle for innovation and reported several directions, including the effectiveness of IDPT, implementation paradigm, and future works.

However, none of the studies reported a reference model of an adaptive IDPT system with its major components nor provided comprehensive guidelines on developing an adaptive IDPT system.

#### **X. CONCLUSION**

In this paper, we propose a model for adaptive IDPT based on classical control theory and discuss how the model can be used to represent an adaptive IDPT system. In addition to the model, we outline the core components of adaptive IDPT systems, the main adaptive elements, dimensions of adaptiveness, information architecture applied to adaptive systems and strategies used in the adaptation process. We also provide comprehensive guidelines on how to develop an adaptive IDPT system building on the Person-Based Approach. Patient adoption of and persistence with the IDPT intervention program is a prerequisite to successful outcomes of the therapy, which requires the combination of the perspectives of computer science and psychology/behaviour science when building the intervention. Here, we have proposed to combine a user-centered design approach with systems that adapt to patient behavior. Adaptive IDPT systems have the potential to offer personalized treatment to the far-reaching population with lesser use of resources, ensuring higher user adherence, reduced dropouts, and better treatment outcomes, and equally important: to achieve successful therapeutic outcomes for those affected.

In the future, we need tools, frameworks, architecture for an adaptive IDPT, and Domain-Specific Languages (DSL) for adaptive IDPT to facilitate the development of scalable, interoperable, and secure adaptive IDPT systems. We also need better dashboard tools that help therapists and other medical practitioners to comprehend the patients status better and adapt their interventions based on their engagement with the interventions. Thoroughly testing the adaptive IDPT system model and all the types of adaptive strategies have been kept as one of our project's immediate work.

#### ACKNOWLEDGMENT

INTROMAT is a Norwegian research and development project that investigates how to develop adaptive technology for meeting mental health challenges. The opinions, findings, discussions, recommendations, and conclusions illustrated in this chapter are those of the authors and do not reflect the views of the funding agencies.

#### REFERENCES

- L. Northrop, P. Feiler, R. P. Gabriel, J. Goodenough, and E. Al, "Ultralarge-scale systems—The software challenge of the future," Softw. Eng. Inst., Carnegie Mellon Univ., Pittsburgh, PA, USA, Tech. Rep., 2006.
- [2] K. Narendra, "Adaptation and learning in automatic systems," *IEEE Trans. Autom. Control*, vol. AC-19, no. 2, pp. 173–174, Apr. 1974.

- [3] S. K. Mukhiya, F. Rabbi, K. I. Pun, and Y. Lamo, "An architectural design for self-reporting e-health systems," in *Proc. IEEE/ACM 1st Int. Workshop Softw. Eng. Healthcare (SEH)*, May 2019, pp. 1–8.
- [4] L. Yardley, L. Morrison, K. Bradbury, and I. Muller, "The person-based approach to intervention development: Application to digital health-related behavior change interventions," *J. Med. Internet Res.*, vol. 17, no. 1, p. e30, Jan. 2015.
- [5] K. Bradbury, K. Morton, R. Band, A. van Woezik, R. Grist, R. J. Mcmanus, P. Little, and L. Yardley, "Using the person-based approach to optimise a digital intervention for the management of hypertension," *PLoS ONE*, vol. 13, no. 5, May 2018, Art. no. e0196868.
- [6] C. Abras, D. Maloney-Krichmar, and J. Preece, "User-centred design," in *Encyclopedia of Human-Computer Interaction*, W. Bainbridge, ed. Thousand Oaks, CA, USA: Sage, 2004, pp. 763–768,.
- [7] L. Morrison, I. Muller, L. Yardley, and K. Bradbury, "The personbased approach to planning, optimising, evaluating and implementing behavioural health interventions," *Eur. Health Psychologist*, vol. 20, no. 3, pp. 464–469, 2018.
- [8] B. H. Cheng, "Software engineering for self-adaptive systems: A research roadmap," in *Software Engineering for Self-Adaptive Systems* (Lecture Notes in Computer Science), vol. 5525. Berlin, Germany: Springer, 2009, pp. 1–26. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-642-02161-9\_1
- [9] J. Whittle, P. Sawyer, N. Bencomo, B. H. C. Cheng, and J.-M. Bruel, "RELAX: Incorporating uncertainty into the specification of self-adaptive systems," in *Proc. 17th IEEE Int. Requirements Eng. Conf.*, Aug. 2009, pp. 79–88.
- [10] P. Oreizy, M. M. Gorlick, R. N. Taylor, D. Heimhigner, G. Johnson, N. Medvidovic, A. Quilici, D. S. Rosenblum, and A. L. Wolf, "An architecture-based approach to self-adaptive software," *IEEE Intell. Syst.*, vol. 14, no. 3, pp. 54–62, May 1999.
- [11] K. De Jong, "Adaptive system design: A genetic approach," *IEEE Trans. Syst., Man, Cybern.*, vol. SMC-10, no. 9, pp. 566–574, Oct. 1980.
- [12] B. G. Danaher, H. Brendryen, J. R. Seeley, M. S. Tyler, and T. Woolley, "From black box to toolbox: Outlining device functionality, engagement activities, and the pervasive information architecture of mHealth interventions," *Internet Interventions*, vol. 2, no. 1, pp. 91–101, Mar. 2015.
- [13] A. Lange, D. Rietdijk, M. Hudcovicova, J.-P. van de Ven, B. Schrieken, and P. M. G. Emmelkamp, "Interapy: A controlled randomized trial of the standardized treatment of posttraumatic stress through the Internet," *J. Consulting Clin. Psychol.*, vol. 71, no. 5, pp. 901–909, 2003.
- [14] B. Meyer, T. Berger, F. Caspar, C. G. Beevers, G. Andersson, and M. Weiss, "Effectiveness of a novel integrative online treatment for depression (Deprexis): Randomized controlled trial," *J. Med. Internet Res.*, vol. 11, no. 2, p. e15, May 2009.
- [15] P. van de Ven, H. O'Brien, R. Henriques, M. Klein, R. Msetfi, J. Nelson, A. Rocha, J. Ruwaard, D. O'Sullivan, and H. Riper, "ULTEMAT: A mobile framework for smart ecological momentary assessments and interventions," *Internet Interventions*, vol. 9, pp. 74–81, Sep. 2017.
- [16] V. Pejovic and M. Musolesi, "Anticipatory mobile computing for behaviour change interventions," in *Proc. ACM Int. Joint Conf. Pervasive Ubiquitous Comput.*, 2014, pp. 1025–1034.
- [17] T. Donker, K. Petrie, J. Proudfoot, J. Clarke, M. R. Birch, and H. Christensen, "Smartphones for smarter delivery of mental health programs: A systematic review," *J. Med. Internet Res.*, vol. 15, no. 11, p. e247, 2013.
- [18] G. Andersson, N. Titov, B. F. Dear, A. Rozental, and P. Carlbring, "Internet-delivered psychological treatments: From innovation to implementation," *World Psychiatry*, vol. 18, no. 1, pp. 20–28, Feb. 2019.
- [19] K. S. Dobson, Handbook Cognitive-Behavioral Therapies. London, U.K.: The Guilford Press, 2010.
- [20] R. Kohn, S. Saxena, I. Levav, and B. Saraceno, "The treatment gap in mental health care," *Bulletin World Health Org.*, vol. 82, pp. 858–866, Nov. 2004.
- [21] W. H. O. 2003, "Adherence to long-term therapies: Evidence for action," in Adherence to Long-Term Therapies: Evidence for Action. Geneva, Switzerland: World Health Organization, 2003, p. 211.
- [22] I. M. Marks, K. Cavanagh, and L. Gega, "Computer-aided psychotherapy: Revolution or bubble?" *Brit. J. Psychiatry*, vol. 191, pp. 471–473, 2007, doi: 10.1192/bjp.bp.107.041152.

- [23] E. E. K. Wallin, S. Mattsson, and E. M. G. Olsson, "The preference for Internet-based psychological interventions by individuals without past or current use of mental health treatment delivered online: A survey study with mixed-methods analysis," *JMIR Mental Health*, vol. 3, no. 2, p. e25, Jun. 2016.
- [24] D. B. Portnoy, L. A. J. Scott-Sheldon, B. T. Johnson, and M. P. Carey, "Computer-delivered interventions for health promotion and behavioral risk reduction: A meta-analysis of 75 randomized controlled trials, 1988–2007," *Preventive Med.*, vol. 47, no. 1, pp. 3–16, Jul. 2008.
- [25] Y. Inal, J. D. Wake, F. Guribye, and T. Nordgreen, "Usability evaluations of mobile mental health technologies: Systematic review," *J. Med. Internet Res.*, vol. 22, no. 1, Jan. 2020, Art. no. e15337.
- [26] S. Nobis, D. Lehr, D. D. Ebert, M. Berking, E. Heber, H. Baumeister, A. Becker, F. Snoek, and H. Riper, "Efficacy and cost-effectiveness of a Web-based intervention with mobile phone support to treat depressive symptoms in adults with diabetes mellitus type 1 and type 2: Design of a randomised controlled trial," *BMC Psychiatry*, vol. 13, no. 1, p. 306, Dec. 2013.
- [27] H. Christensen, K. M. Griffiths, A. J. Mackinnon, and K. Brittliffe, "Online randomized controlled trial of brief and full cognitive behaviour therapy for depression," *Psychol. Med.*, vol. 36, no. 12, pp. 1737–1746, Dec. 2006.
- [28] H. Baumeister, L. Reichler, M. Munzinger, and J. Lin, "The impact of guidance on Internet-based mental health interventions—A systematic review," *Internet Interventions*, vol. 1, no. 4, pp. 205–215, Oct. 2014.
- [29] G. Andersson and N. Titov, "Advantages and limitations of Internet-based interventions for common mental disorders," *World Psychiatry*, vol. 13, no. 1, pp. 4–11, Feb. 2014.
- [30] D. C. Mohr, P. Cuijpers, and K. Lehman, "Supportive accountability: A model for providing human support to enhance adherence to eHealth interventions," *J. Med. Internet Res.*, vol. 13, no. 1, p. e30, Mar. 2011.
- [31] M. L. Tielman, M. A. Neerincx, M. van Meggelen, I. Franken, and W.-P. Brinkman, "How should a virtual agent present psychoeducation? Influence of verbal and textual presentation on adherence," *Technol. Health Care*, vol. 25, no. 6, pp. 1081–1096, Feb. 2018.
- [32] N. Titov, B. F. Dear, L. Johnston, C. Lorian, J. Zou, B. Wootton, J. Spence, P. M. McEvoy, and R. M. Rapee, "Improving adherence and clinical outcomes in self-guided Internet treatment for anxiety and depression: Randomised controlled trial," *PLoS ONE*, vol. 8, no. 7, Jul. 2013, Art. no. e62873.
- [33] J. Boettcher, T. Berger, and B. Renneberg, "Does a pre-treatment diagnostic interview affect the outcome of Internet-based self-help for social anxiety disorder? A randomized controlled trial," *Behavioural Cognit. Psychotherapy*, vol. 40, no. 5, pp. 513–528, Oct. 2012.
- [34] M. Flores, G. Glusman, K. Brogaard, N. D. Price, and L. Hood, "P4 medicine: How systems medicine will transform the healthcare sector and society," *Personalized Med.*, vol. 10, no. 6, pp. 565–576, Aug. 2013.
- [35] B. S. Fernandes, L. M. Williams, J. Steiner, M. Leboyer, A. F. Carvalho, and M. Berk, "The new field of 'precision psychiatry," *BMC Med.*, vol. 15, p. 80, Apr. 2017.
- [36] J. J. Goldberger and A. E. Buxton, "Personalized medicine vs guidelinebased medicine," *JAMA*, vol. 309, pp. 2559–2560, Jun. 2013.
- [37] K. A. Hallgren, A. M. Bauer, and D. C. Atkins, "Digital technology and clinical decision making in depression treatment: Current findings and future opportunities: HALLGREN," *Depression Anxiety*, vol. 34, no. 6, pp. 494–501, Jun. 2017.
- [38] P. Bower and S. Gilbody, "Stepped care in psychological therapies: Access, effectiveness and efficiency. Narrative literature review.," *The Brit. J. Psychiatry, J. Mental Sci.*, vol. 186, pp. 7–11, Jan. 2005.
- [39] J. Andersson, R. De Lemos, S. Malek, and D. Weyns, "Modeling dimensions of self-adaptive software systems," in *Software Engineering for Self-Adaptive Systems* (Lecture Notes in Computer Science), vol. 5525. Berlin, Germany: Springer, 2009, pp. 27–47. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-642-02161-9\_2
- [40] C. Liu, R. W. White, and S. Dumais, "Understanding Web browsing behaviors through weibull analysis of dwell time," in *Proc.33rd Int. ACM SIGIR Conf. Res. Develop. Inf. Retr.*, 2010, pp. 379–386.
- [41] M. R. Scholten, S. M. Kelders, and J. E. Van Gemert-Pijnen, "Self-guided Web-based interventions: Scoping review on user needs and the potential of embodied conversational agents to address them," *J. Med. Internet Res.*, vol. 19, no. 11, p. e383, Nov. 2017, doi: 10.2196/jmir.7351.

- [42] N. Saeed, Y. Yang, and S. Sinnappan, "Emerging Web technologies in higher education: A case of incorporating blogs, podcasts and social bookmarks in a Web programming course based on students' learning styles and technology preferences," *Educ. Technol. Soc.*, vol. 12, pp. 98–109, 10 2009.
- [43] M. Agosti and G. M. Di Nunzio, "Gathering and mining information from Web log files," in *Digital Libraries: Research* and Development (Lecture Notes in Computer Science), vol. 4877. Berlin, Germany: Springer, 2007, pp. 104–113. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-540-77088-6\_10
- [44] S. Gilbody, D. Richards, S. Brealey, and C. Hewitt, "Screening for depression in medical settings with the patient health questionnaire (PHQ): A diagnostic meta-analysis," *J. Gen. Internal Med.*, vol. 22, no. 11, pp. 1596–1602, Oct. 2007.
- [45] L. C. Quilty, J. J. Robinson, J. P. Rolland, F. D. Fruyt, F. Rouillon, and R. M. Bagby, "The structure of the Montgomery-Asberg depression rating scale over the course of treatment for depression," *Int. J. Methods Psychiatric Res.*, vol. 22, pp. 175–184, Sep. 2013.
- [46] A. T. Beck, R. A. Steer, and M. G. Carbin, "Psychometric properties of the beck depression inventory: Twenty-five years of evaluation," *Clin. Psychol. Rev.*, vol. 8, no. 1, pp. 77–100, Jan. 1988.
- [47] R. L. Spitzer, "A brief measure for assessing generalized anxiety disorder: The GAD-7," Arch. Internal Med., vol. 166, no. 10, pp. 1092–1097, 2006.
- [48] A. T. Chen, S. Wu, K. N. Tomasino, E. G. Lattie, and D. C. Mohr, "A multifaceted approach to characterizing user behavior and experience in a digital mental health intervention," *J. Biomed. Informat.*, vol. 94, Jun. 2019, Art. no. 103187.
- [49] K. Ramachandran, "Adaptive user interfaces for health care applications," in *Proc. IBM Developerworks*, Jan. 2009, pp. 1–8.
- [50] J. Pugatch, E. Grenen, S. Surla, M. Schwarz, and H. Cole-Lewis, "Information architecture of Web-based interventions to improve health outcomes: Systematic review," *J. Med. Internet Res.*, vol. 20, no. 3, p. e97, Mar. 2018.
- [51] R. Crutzen, D. Cyr, and N. K. De Vries, "The role of user control in adherence to and knowledge gained from a Website: Randomized comparison between a tunneled version and a freedom-of-choice version," *J. Med. Internet Res.*, vol. 14, no. 2, pp. 75–84, 2012.
- [52] P. Morville, and L. Rosenfeld, *Information Architecture for the World Wide Web: Designing Large-Scale Web Sites*, 3rd ed. Sebastopol, CA, USA: O'Reilly Media, 2006.
- [53] C. K. Danielson, J. L. McCauley, K. S. Gros, A. M. Jones, S. C. Barr, A. L. Borkman, B. G. Bryant, and K. J. Ruggiero, "SiHLEWeb.Com: Development and usability testing of an evidence-based HIV prevention Website for female african-american adolescents," *Health Informat. J.*, vol. 22, no. 2, pp. 194–208, Jun. 2016.
- [54] J. J. Garrett, Elements User Experience, The: User-Centered Design for Web Beyond. London, U.K.: Pearson, 2010.
- [55] S. M. Kelders, R. N. Kok, H. C. Ossebaard, and J. E. Van Gemert-Pijnen, "Persuasive system design does matter: A systematic review of adherence to Web-based interventions," *J. Med. Internet Res.*, vol. 14, no. 6, p. e152, 2012. [Online]. Available: https://www.jmir.org/2012/6/e152, doi: 10.2196/jmir.2104.
- [56] P. J. Lynch and S. Horton, Web Style Guide: Foundations of User Experience Design. New Haven, CT, USA: Yale Univ. Press, 2016.
- [57] B. G. Danaher, H. G. McKay, and J. R. Seeley, "The information architecture of behavior change websites," *J. Med. Internet Res.*, vol. 7, no. 2, p. e12, 2005.
- [58] V. I. Voloshin, *Introduction to Graph Theory*. Hauppauge, NY, USA: Nova Science, 2009.
- [59] E. Garcia-Ceja, M. Riegler, T. Nordgreen, P. Jakobsen, K. J. Oedegaard, and J. Tørresen, "Mental health monitoring with multimodal sensing and machine learning: A survey," *Pervasive Mobile Comput.*, vol. 51, pp. 1–26, 2018.
- [60] C. Gobert, K. Todi, G. Bailly, and A. Oulasvirta, "SAM: A modular framework for self-adapting Web menus," in *Proc. 24th Int. Conf. Intell. User Interfaces*, Mar. 2019, pp. 481–485.
- [61] M. Salehie and L. Tahvildari, "Self-adaptive software: Landscape and research challenges," ACM Trans. Auto. Adapt. Syst., vol. 4, no. 2, pp. 1–42, May 2009.
- [62] N. A. Qureshi, A. Perini, N. A. Ernst, and J. Mylopoulos, "Towards a continuous requirements engineering framework for self-adaptive systems," in *Proc. 1st Int. Workshop Requirement*, Sep. 2010, pp. 9–16.

- [63] L. Sabatucci, V. Seidita, and M. Cossentino, "The four types of selfadaptive systems: A metamodel," in *Smart Innovation, Systems and Technologies*, vol. 76. Cham, Switzerland: Springer, 2018, pp. 440–450. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-319-59480-4\_44
- [64] A. Horn, "On sentences which are true of direct unions of algebras," J. Symbolic Log., vol. 16, pp. 14–21, Dec. 1951.
- [65] L. Jokste and J. Grabis, "Rule based adaptation: Literature review," in *Vide. Tehnologija. Resursi—Environment, Technology, Resources*, vol. 2. Rēzekne, Latvia: Rezekne Higher Education Institution, 2017, pp. 42–46.
- [66] E. Peukert, J. Eberius, and E. Rahm, "Rule-based construction of matching processes," in *Proc. 20th ACM Int. Conf. Inf. Knowl. Manage. (CIKM)*, 2011, pp. 2421–2424.
- [67] T. Zhao, H. Zhao, W. Zhang, and Z. Jin, "User preference based autonomic generation of self-adaptive rules," in *Proc. 6th Asia–Pacific Symp. Internetware*, 2014, pp. 25–34.
- [68] V. López-Jaquero, F. Montero, and F. Real, "Designing user interface adaptation rules with T: XML," in *Proc. Proceedingsc 13th Int. Conf. Intell. user Interface (IUI)*, 2008, pp. 383–387.
- [69] L. Canning and S. H. Lloyd, "Modelling the adaptation process in interactive business relationships," J. Bus. Ind. Marketing, vol. 17, no. 7, pp. 615–636, Dec. 2002.
- [70] J. He, T. Gao, W. Hao, I.-L. Yen, and F. Bastani, "A flexible content adaptation system using a rule-based approach," *IEEE Trans. Knowl. Data Eng.*, vol. 19, no. 1, pp. 127–140, Jan. 2007.
- [71] Y. Brun, G. Di Marzo Serugendo, C. Gacek, H. Giese, H. Kienle, M. Litoiu, H. Müller, M. Pezzè, and M. Shaw, "Engineering selfadaptive systems through feedback loops," in *Software Engineering for Self-Adaptive Systems* (Lecture Notes in Computer Science), vol. 5525, B. H. C. Cheng, R. de Lemos, H. Giese, P. Inverardi, and J. Magee, Eds. Berlin, Germany: Springer-Verlag, 2009, pp. 48–70.
- [72] S. Dobson, S. Denazis, A. Fernández, D. Gaïti, E. Gelenbe, F. Massacci, P. Nixon, F. Saffre, N. Schmidt, and F. Zambonelli, "A survey of autonomic communications," *ACM Trans. Auton. Adapt. Syst.*, vol. 1, no. 2, pp. 223–259, 2006.
- [73] W. Heaven, D. Sykes, J. Magee, and J. Kramer, "A case study in goal-driven architectural adaptation," in *Software Engineering for Self-Adaptive Systems* (Lecture Notes in Computer Science), vol. 5525. Berlin, Germany: Springer, 2009, pp. 109–127. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-642-02161-9\_6
- [74] J. Wallert, E. Gustafson, C. Held, G. Madison, F. Norlund, L. Von Essen, and E. M. G. Olsson, "Predicting adherence to Internet-Delivered psychotherapy for symptoms of depression and anxiety after myocardial infarction: Machine learning insights from the U-CARE heart randomized controlled trial," *J. Med. Internet Res.*, vol. 20, Oct. 2018, Art. no. e10754.
- [75] S. K. Mukhiya, F. Rabbiab, V. K. I. Punax, A. Rutle, and Y. Lamo, "A Graphql approach to healthcare information exchange with HL7 FHIR," *Procedia Comput. Sci.*, vol. 160, pp. 338–345, Jun. 2019.
- [76] I. Sommerville, Software Engineering, 10th ed. London, U.K.: Pearson, 2016. [Online]. Available: https://www.pearson.com/us/highereducation/program/Sommerville-Software-Engineering-10th-Edition/PGM35255.html
- [77] IEEE Guide to Software Requirements Specifications, Standard 830-1984, 1984.
- [78] B. Rumpe and B. Rumpe, "Object Constraint Language," in *Proc. Modeling UML*. Cham, Switzerland: Springer, 2016, pp. 37–101. [Online]. Available: https://www.springer.com/gp/book/9783319339320
- [79] A. M. Langer, Guide to Software Development Designing and Managing the Life Cycle. London, U.K.: Springer-Verlag, 2016.
- [80] S. Kumar, W. Nilsen, M. Pavel, and M. Srivastava, "Mobile health: Revolutionizing healthcare through transdisciplinary research," *Computer*, vol. 46, no. 1, pp. 28–35, Jan. 2013.
- [81] P. Liao, P. Klasnja, A. Tewari, and S. A. Murphy, "Micro-randomized trials in mHealth," *Statist. Med.*, vol. 35, no. 12, pp. 1944–1971, Apr. 2015.
- [82] T. L. Webb, J. Joseph, L. Yardley, and S. Michie, "Using the Internet to promote health behavior change: A systematic review and meta-analysis of the impact of theoretical basis, use of behavior change techniques, and mode of delivery on efficacy," *J. Med. Internet Res.*, vol. 12, no. 1, p. e4, Feb. 2010.
- [83] H. Christensen, K. M. Griffiths, and L. Farrer, "Adherence in Internet interventions for anxiety and depression: Systematic review," J. Med. Internet Res., vol. 11, no. 2, p. e13, 2009.

- [84] J. Kelson, A. Rollin, B. Ridout, and A. Campbell, "Internet-delivered acceptance and commitment therapy for anxiety treatment: Systematic review," *J. Med. Internet Res.*, vol. 21, no. 1, p. e12530, 2019.
- [85] E. Karyotaki, "Predictors of treatment dropout in self-guided Web-based interventions for depression: An 'individual patient data' meta-analysis," *Psychol. Med.*, vol. 45, pp. 2717–2726, Oct. 2015.
- [86] M. A. Rogers, K. Lemmen, R. Kramer, J. Mann, and V. Chopra, "Internetdelivered health interventions that work: Systematic review of metaanalyses and evaluation of website availability," *J. Med. Internet Res.*, vol. 19, no. 3, p. e90, 2017.
- [87] W. Brouwer, W. Kroeze, R. Crutzen, J. de Nooijer, N. K. de Vries, J. Brug, and A. Oenema, "Which intervention characteristics are related to more exposure to Internet-delivered healthy lifestyle promotion interventions? A systematic review," J. Med. Internet Res., vol. 13, no. 1, p. e2, Jan. 2011.



**JO DUGSTAD WAKE** received the Ph.D. degree in information science from the Department of Information Science and Media Studies, in 2013. The Ph.D. was on developing, implementing, and evaluation of mobile and location-based games for learning. He is currently a Senior Scientist with the NORCE Norwegian Research Centre, Bergen, Norway. His current research interests include human-computer interaction and interaction design and analysis, digital health and health

informatics, computer supported collaborative learning, mobile learning, and game-based learning.



**YAVUZ INAL** received the Ph.D. degree from Middle East Technical University, Turkey. The Ph.D. was on designing and developing physically interactive serious games for children. He is currently a Postdoctoral Fellow with the Department of Information Science and Media Studies, University of Bergen, Norway. His research interests include human–computer interaction, interaction design, usability evaluation, user experience, digital accessibility, game studies, player experience, information systems, and health informatics.



**SURESH KUMAR MUKHIYA** (Member, IEEE) received the bachelor's degree in computer science and information technology from Tribhuvan University, Nepal, and the master's degree in information systems from the Norwegian University of Science and Technology (NTNU), Trondheim, Norway. He is currently pursuing the Ph.D. degree with the Western Norway University of Applied Sciences, Bergen, Norway. His research interests include model-based software engineering.

front-end development, data science, and health informatics. He was awarded the Vice Chancellor Award by Tribhuvan University.



**YNGVE LAMO** received the Ph.D. degree in computer science from the University of Bergen, on formal specification of software systems with use of multialgebras, in 2003. He is currently a Professor with the Department of Computer Science, Electrical Engineering and Mathematical Sciences, Western Norway University of Applied Sciences, Bergen, Norway. His research interests include model-based software engineering, formal methods, graph transformations, and health informatics.

. . .